Author’s response to reviews

Title: A case of hypertrophic cardiomyopathy combined with muscular ventricular septal defect and abnormal origin of right coronary artery

Authors:

Guang-mei Zheng (zgm_4m@163.com)
Jiao Bai (baijiao2007@yahoo.com.cn)
Jun-ming Tang (tangjm416@163.com)
Fang-cheng Zhu (474853596@qq.com;zfc_8m@163.com)
Hong-xia Jing (20351424@qq.com)

Version: 4 Date: 25 Dec 2018

Author’s response to reviews:

Dear Editor:

Thank you very much for your letter and constructive comments. The manuscript has been revised according to the comments. We hope that the revised manuscript is acceptable for publishing in BMC Cardiovascular Disorders.

Sincerely yours,

Fang-cheng Zhu

Department of Forensic Medicine, Hubei University of Medicine,
Shiyan 442000, Hubei China
Tel: +86 719 8637170;
Email: 474853596@qq.com
Editor Comments:

Thank you for amending your manuscript according to our requests. However, there are a few minor revisions that are necessary before we can accept the manuscript for publication:

1. Overlap:
We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the Discussion and Conclusions sections.
While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.
Please re-phrase these sections to minimise overlap.

Response: Thank you for your constructive comments. We have corrected the overlap mistakes in the part of discussion (Discussion and Conclusions sections, line 10-12, paragraph 3, page )

2. Section:
Upon submission of your revised manuscript, please assign to 'Coronary Artery Disease'.
Response: Thanks. We have done it as you suggested.